Author's response to reviews

Title: Factors influencing the consumption of seafood among young children in Perth: a qualitative study

Authors:

Alexandra McManus Dr (A.Mcmanus@curtin.edu.au)
Sharyn K Burns (S.Burns@curtin.edu.au)
Peter A Howat Professor (P.Howat@curtin.edu.au)
Lisa Cooper (L.Cooper@curtin.edu.au)
Lynda Fielder (L.Fielder@curtin.edu.au)

Version: 6 Date: 1 June 2007

Author's response to reviews:

Dear Shinese,

Attached is the completed manuscript as requested. I hope this is appropriate.

The following has been added to the previous manuscript:
- appropriate formatting;
- a statement confirming ethics approval was received and this ethics approval complies with the Helsinki Declaration,
- written consent was received prior to participants being involved in the study;
- all authors emails are included; and
- author contributions and competing interests statements are listed.

Thank you for your support in progressing this manuscript to print.

Kind regards
Alexandra